

PEER REVIEW HISTORY

BMJ Open publishes all reviews undertaken for accepted manuscripts. Reviewers are asked to complete a checklist review form (<http://bmjopen.bmj.com/site/about/resources/checklist.pdf>) and are provided with free text boxes to elaborate on their assessment. These free text comments are reproduced below.

ARTICLE DETAILS

| | |
|----------------------------|---|
| TITLE (PROVISIONAL) | International Phase I Study Protocol to Develop a Patient-Reported Outcome Measure for Adolescents and Adults Receiving Gender-Affirming Treatments (the GENDER-Q) |
| AUTHORS | Klassen, Anne; Kaur, Manraj; Johnson, Natasha; Kreukels, Baudewijntje PC; McEvenue, Giancarlo; Morrison, Shane; Mullender, Margriet G; Ozer, Mujde; Rowe, Will; Satterwhite, Thomas; Savard, Kinusan; Semple, John; van de Grift, Tim C; van der Meij-Ross, Maeghan; Young-Afat, Danny; Pusic, Andrea L |

This article was not externally reviewed at BMJ Open. Protocols that have been independently assessed prior to submission to BMJ Open are usually fast-tracked to publication on the grounds that further substantial changes will not be possible. This independent assessment will usually be external, independent review for both a substantial grant award from a non-commercial or government funder and ethics approval.